Author's response to reviews

Title: Co-morbidity and drug treatment in Alzheimer's disease. A cross sectional study of participants in the Dementia Study in Northern Norway.

Authors:

Fred Andersen (frea@online.no)
Matti Viitanen (matti.viitanen@ki.se)
Dag S Halvorsen (dag.seeger.halvorsen@unn.no)
Bjørn Straume (bjorn.straume@ism.uit.no)
Torgeir A Engstad (torgeir.a.engstad@unn.no)

Version: 4 Date: 22 August 2011

Author's response to reviews: see over
Dear Editor

Thank you for accepting our paper “Co-morbidity and drug treatment in Alzheimer’s disease. A cross sectional study of participants in The Dementia Study in Northern Norway” (BMC Manuscript ID 1425928854549845).

The manuscript has now been formatted and proof read according to your request.

Major revision:
1. The affiliation for each author is written in full according to your template.
2. I confirm that the last version of the abstract have been uploaded
3. Trial registration is inserted as the last section of the abstract with registration site and the unique identifying number.

Minor Revision:
1. Underlining is removed.
2. Abbreviations are listed in a paragraph in alphabetical order and spelled out as they occur in the literature.
3. All authors are now named in the reference list and “et al” is deleted.
4. The lay-out of the tables is revised according to your request.
5. “White space” in and around the figure is minimized.

We have proof read the manuscript and revised a few typographical error.

We hope that this final revision of the manuscript is in accordance with your expectations. Thank you for publishing our paper.

Yours sincerely

Fred Andersen